Author’s response to reviews

Title: Transcriptome profiling reveals significant changes in the gastric muscularis externa with obesity that partially overlap those that occur with idiopathic gastroparesis

Authors:

Brian Herring (pherring@iupui.edu)
Meng Chen (mchen16@mdanderson.org)
Plamen Mihaylov (pmihaylo@iupui.edu)
April Hoggatt (ahoggatt@iupui.edu)
Anita Gupta (anigupta@iu.edu)
Attila Nakeeb (anakeeb@iupui.edu)
Jennifer Choi (jenchoi@iupui.edu)
John Wo (jmwo@iu.edu)

Version: 2 Date: 04 Jun 2019

Author’s response to reviews:

Updates have been made to the declarations section to meet editorial guidelines.